Author's response to reviews

Title: Comparative Effectiveness of Psychological Treatments for Depressive Disorders in Primary Care: Network Meta-Analysis

Authors:

Klaus Linde (Klaus.Linde@tum.de)
Gerta Rücker (ruecker@imbi.uni-freiburg.de)
Kirsten Sigterman (kirsten@sigterman.nl)
Susanne Jamil (susanneohnsmann@gmx.de)
Karin Meissner (Karin.Meissner@med.uni-muenchen.de)
Antonius Schneider (antoniusschneider@tum.de)
Levente Kriston (l.kriston@uke.uni-hamburg.de)

Version: 2
Date: 14 January 2015

Author's response to reviews:

Thank you for your careful manuscript check.
Please find our point by point response to the request s below.

1. Title page. It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
R: The missing Email addresses of co-authors have been added; the postal address of the corresponding author has been added in the respective section. The other information had also been included in our original submission.

2. Line Numbering
R: has been added, page breaks were removed.

3. Ethics
R: Ethical approval was not necessary (systematic review of published research). The PRISMA Checklist for reporting systematic reviews has been filled in and is submitted now in an additional file.

4. Availability of Supporting Data section
We added an availability of supporting data statement stating that the raw data used in is included in the online supplemental material section 4.

5. Please move your Author's Contribution section after Acknowledgement
R: The Author's Contribution section is placed after Acknowledgement.